Author's response to reviews

Title: Regional differences in multidimensional aspects of health: findings from the MRC Cognitive Function and Ageing study

Authors:

Fiona E Matthews (fiona.matthews@mrc-bsu.cam.ac.uk)
Laura Miller (L.L.Miller@bristol.ac.uk)
Carol Brayne (carol.brayne@phpc.cam.ac.uk)
Carol Jagger (cxj@le.ac.uk)
Mrc Cfas (fiona.matthews@mrc-bsu.cam.ac.uk)

Version: 2 Date: 24 January 2006

Author's response to reviews:

Revised the conflict of interest statement.